Author's response to reviews

Title: Prevalence And Modes Of Complementary And Alternative Medicine Use Among Peasant Farmers With Musculoskeletal Pain In A Rural Community In South-Western Nigeria

Authors:

Chidozie E Mbada (doziembada@yahoo.com)
Tijani L Adeyemi (lukmanadeyemi@yahoo.com)
Rufus A Adedoyin (radedoyi@yahoo.com)
Hakeem D Badmus (hbadmus@yahoo.com)
Taofeek O Awotidebe (tidebet@yahoo.com)
Olujide O Arije (olujide_arije@yahoo.com)
Olorunfemi S Omotosho (davfemsinc@yahoo.com)

Version: 6 Date: 20 May 2015

Author's response to reviews:

The Editor in Chief,
Journal: BMC Complementary and Alternative Medicine

Dear Sir,

Correction to study’s questionnaire - Prevalence And Modes Of Complementary And Alternative Medicine Use Among Peasant Farmers With Musculoskeletal Pain In A Rural Community In South-Western Nigeria

Thank you for the editorial comments on the questionnaire used in this study. I hereby state that the original questionnaire without rewording changes with respect to definition of CAM and mind-body therapy respectively is attached. The differences in the present attachment and the previously submitted questionnaire is presented in bold format.

Thank you immensely,

Chidozie E. Mbada PhD.